Author's response to reviews

Title: Incidence, Mortality Patterns and 5-year Relative Survival Ratio of Prostate Cancer among Residents in Singapore from 1968 to 2002

Authors:

Sin Eng Chia (cofcse@nus.edu.sg)
Chuen Seng Tan (cmetcs@nus.edu.sg)
Gek Hsiang Lim (cmelgh@nus.edu.sg)
Xueling Sim (cmesx@nus.edu.sg)
Yudi Pawitan (Yudi.Pawitan@ki.se)
Marie Reilly (Marie.Reilly@ki.se)
Safiyya Mohamed Ali (safiyya@nus.edu.sg)
Kee Seng Chia (cofcks@nus.edu.sg)

Version: 2 Date: 25 March 2008

Author's response to reviews:

Added cover letter